Author’s response to reviews

Title: Chronic diarrhea associated with persistent norovirus excretion in patients with chronic lymphocytic leukemia: report of two cases

Authors:

Todd Capizzi (tcapizzi@denvercancercare.com)
Grace Makari-Judson (grace.makari-judson@baystatehealth.org)
Richard Steingart (richard.steingart@baystatehealth.org)
Wilson C Mertens (wilson.mertens@baystatehealth.org)

Version: 5  Date: 12 May 2011

Author’s response to reviews:

I tried to post this weeks ago, but cannot find the location on the website to place a revised-revised manuscript. The final is attached to this email, and I will try again.

To the extent possible, allowing for grammar, all requests have been fulfilled.

-----Original Message-----
From: The BMC Series Editorial Production Team [mailto:bmcseriesprepublication@biomedcentral.com]
Sent: Thursday, May 12, 2011 9:30 AM
To: Mertens, Wilson
Cc: Todd Capizzi; Makari-Judson, Grace; Steingart, Richard; Mertens, Wilson
Subject: Journal: BMC Infectious Diseases, MS:2030060454433130, Further Formatting changes required: Due 14/05/2011

Author list: Todd Capizzi, Grace Makari-Judson, Richard Steingart and Wilson C Mertens
Title : Chronic diarrhea associated with persistent norovirus excretion in patients with chronic lymphocytic leukemia: report of two cases
Journal: BMC Infectious Diseases
MS ID : 2030060454433130

Dear Dr Mertens,

We are still waiting for you to provide a revised submission to BMC Infectious Diseases. In order to minimise delay we have contacted all available authors.

We may have edited the manuscript for you, please ensure that all your revisions are made to the most recent version of your manuscript, which can be accessed at the following link:
http://www.biomedcentral.com/media/man/2030060454433130/msd/main/0

Please ensure that all requested formatting changes are made and that you proof-read your manuscript for errors before you resubmit it. Instructions on how to submit your revised manuscript are available below.

Please note that this is the final proofing stage. All revised manuscripts submitted to BMC Infectious Diseases are regarded as being ready for publication. After publication, you will not be able to make any corrections to errors present in this version. Please check your manuscript carefully for any typing or content errors, as these will be included in the published version.

Please resubmit your manuscript here via the online submission system:
http://www.biomedcentral.com/author/manuscript/details/view.do?manuscriptId=2030060454433130

Formatting changes to be made:
----------------------------

Major revisions (we require the author to make these changes)

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Case Presentation, Conclusions

Structure: this manuscript does not have the correct structure. Please reformat it to use these headings: Abstract; Background; Case Presentation; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Authors’ information (optional); Acknowledgements (optional); References

Authors’ contributions: Please place the Authors’ contributions section towards the end of the manuscript text, directly preceding the Acknowledgements.

Authors’ contributions: Please use the following format as an example, including using initials to refer to each author’s contribution: AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

References: Please make sure references are cited in numerical order. We notice reference 6 is not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited.

Tables: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.
Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please remember to include the country in the affiliation details

References: the reference list should contain up to 30 authors¿ names for each citation. The term ¿et al.¿ should not be used if there are fewer than 30 authors.

Additional files: Please remove the copy of your paper with tracked changes from the system.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Thank you